

This item is the archived peer-reviewed author-version of:

Gender discrimination? Evidence from the Belgian public accounting profession

Reference:

Hardies Kris, Lennox Clive, Li Bing.- Gender discrimination? Evidence from the Belgian public accounting profession Contemporary accounting research - ISSN 0823-9150 - 38:3(2021), 1911-3846.12667 Full text (Publisher's DOI): https://doi.org/10.1111/1911-3846.12667 To cite this reference: https://hdl.handle.net/10067/1745950151162165141

uantwerpen.be

Institutional repository IRUA

Gender Discrimination? Evidence from the Belgian Public Accounting Profession

Kris Hardies University of Antwerp, Belgium kris.hardies@uantwerpen.be

Clive Lennox University of Southern California, United States clennox@marshall.usc.edu

Bing Li City University of Hong Kong, Hong Kong bingli@cityu.edu.hk

Contemporary Accounting Research (forthcoming)

We gratefully acknowledge the comments of Miguel Minutti-Meza (the editor), two anonymous reviewers, Jayanthi Krishnan, Zawadi Lemayian, Chan Li, Jeff Pittman, and workshop participants at George Mason University, Georgia State University, Temple University, the University of Southern California, the University of Pittsburgh, and Washington University at St. Louis.

Gender Discrimination? Evidence from the Belgian Public Accounting Profession

ABSTRACT

Prior research finds that women receive lower salaries than men. Similarly, we show that female audit partners in Belgium receive significantly lower compensation than male partners. However, there are alternative explanations for the pay gap other than gender discrimination. For example, the gap in compensation could reflect that men are paid more because they have higher levels of productivity. We provide new predictions and tests of gender discrimination by comparing the fees generated by audit partners (a measure of partner productivity) and the types of clients assigned to partners. Consistent with female partners having to meet higher performance thresholds than male partners, we show that female partners generate larger fee premiums, but they are less likely to be assigned to prestigious clients. To test whether these patterns are attributable to gender discrimination, we examine whether the results are stronger in maledominated offices because this is where we would expect to find the most discrimination against women. We find the fee premiums generated by female partners are larger in male-dominated offices. Collectively, our combined predictions and tests are consistent with female partners facing gender discrimination in audit offices that are dominated by male partners.

Keywords: gender discrimination; public accounting firms; female partners.

JEL classifications: D22, J71, M41, M42, M51.

1. Introduction

There is a wealth of evidence that women are paid less than men. While this pay gap is consistent with gender discrimination, an alternative explanation is that men are paid more because they have higher levels of productivity (Oaxaca and Ransom 1994; Blau and Kahn 2000, 2017). In this study, we generate novel tests for gender discrimination by examining the public accounting profession, where a partner's productivity can be measured directly by examining the partner's ability to generate fee premiums for their employer.

In our first test, we compare the relative productivity of male and female partners (i.e., their ability to generate fee premiums). If accounting firms discriminate against women, we would expect firms to apply higher performance standards to women than men. This would mean that the average female partner would have to be more productive than the average male partner to attain the position of audit partner. Therefore, we test whether female partners generate significantly larger fees compared with their male counterparts (H1a). In our second test, we compare the types of clients assigned to male and female partners. If accounting firms discriminate against women, we would expect more prestigious clients to be assigned to male partners (H2a).

There could be alternative explanations for why male partners generate lower fees than female partners and why male partners are assigned to more prestigious clients. For instance, it could be argued that female partners generate higher fees because they are more productive, whereas female partners are less likely to be assigned to prestigious clients because such clients prefer their partners to be men. To address these alternative explanations, we combine our predictions in H1a and H2a with a set of cross-sectional tests based on the gender composition of individual accounting offices. Survey-based research suggests that discrimination against women

is more likely to occur in work environments dominated by men (Hultin and Szulkin 1999; Gorman 2005; Cohen and Huffman 2007; Kurtulus and Tomaskovic-Devey 2012; Dalton et al. 2014). Therefore, if gender discrimination explains why female partners generate larger fee premiums than male partners (H1a), we would expect this result to be stronger in the offices that have relatively more male partners. This leads to the prediction that the fee premium to female partners is significantly larger in offices dominated by male partners (H1b). Similarly, if gender discrimination explains the tendency of female partners to be assigned to less prestigious clients (H2a), we would expect this result to be stronger in male-dominated offices. Therefore, we hypothesize that the tendency for less prestigious clients to be assigned to female partners is significantly stronger in offices with a higher percentage of male partners (H2b).

We test these four predictions (H1a, H1b, H2a, H2b) using a panel dataset of 94,882 private and public company audits over the period 2008-2014. We examine Belgium rather than the U.S. for two reasons. First, we have a longitudinal panel of data on audit partners in Belgium whereas the U.S. data on audit partners have only been available since 2017. Second, the Belgian setting provides data on both public and private company audits. This is helpful for our study because small private companies are less prestigious than large publicly traded companies, and we expect non-prestigious clients to be assigned to female partners (H2a) particularly in the offices dominated by male partners (H2b). These hypotheses cannot be easily tested in the U.S. setting due to the lack of data on private company audits. Our sample comprises 94,882 client-year observations, 656 unique partners (100 female and 556 male partners), and 199 audit offices. The proportion of male partners in each office ranges from 0% to 100%, implying meaningful variation in the representation of male and female partners. The mean (median) proportion of

male partners in each office is 88% (92%), reflecting that offices generally have more male than female partners.

We begin by testing whether female partners are more productive than their male counterparts (H1a). We test this by comparing the fees generated by female and male partners. After controlling for client characteristics, audit partner characteristics, audit firm characteristics, city fixed effects, year fixed effects, and industry fixed effects, we find a statistically significant fee premium to female partners (H1a). More importantly, the fee premium to female partners is found to be significantly larger in offices with a higher percentage of male partners (H1b), which is where we would expect the most discrimination. We estimate that the fee premium to female partner percentage to 11.1% in offices with the 75th percentile value of the male partner percentage.¹ When we aggregate the fee premiums across all 9,907 audits of the female partners, we find that the total premium ranges from 4.5 million Euros in offices with the 75th percentile value.

Next, we test our prediction that prestigious clients tend to be assigned to male rather than female partners. Consistent with this prediction, we find that publicly traded or large companies are typically assigned to male partners, whereas private or small companies are usually assigned to female partners (H2a). More importantly, we show that this client assignment pattern is significantly stronger in offices with relatively more male partners (H2b). In other words, female partners fare worse in client assignment decisions when they work in male-

¹ Our primary analysis focuses on audit fees rather than non-audit fees because audit partners are primarily responsible for negotiating audit fees. However, audit partners sometimes refer their clients to non-audit partners which helps to generate additional non-audit fees for the firm. In a supplementary analysis, we find similar results using total fees (i.e., audit and non-audit fees).

dominated offices despite that the female partners in these offices perform particularly well in terms of generating larger fee premiums.

Overall, our four sets of results are consistent with women experiencing significant discrimination when they work in offices dominated by male partners. We argue that it is difficult to attribute these results to intrinsic differences between men and women or differences in labor market conditions across the country. Moreover, our results continue to hold when we match the clients of female and male partners using propensity scores and when we match female partners to male partners based on partners' observable characteristics (e.g., partner experience and specializations).²

In a supplementary analysis, we examine the propensity of female partners to attract new clients to the audit firm. We find that 12.3% of new clients are audited by female partners whereas 10.4% of existing clients are audited by female partners, which suggests female partners perform particularly well in terms of attracting new clients to the audit firm. We also examine how incumbent clients are re-assigned following audit partner rotations. Among the prestigious clients whose former partner was female, we find that 100% of clients are subsequently transferred to a male partner. Among the non-prestigious clients whose former partner was male, we find that 16% are subsequently transferred to a female partner. Therefore, when partners are rotated, the evidence suggests that prestigious clients tend to gravitate towards male partners whereas non-prestigious clients gravitate towards female partners. Nevertheless, we are cautious in drawing strong inferences from these results because there are relatively few prestigious

 $^{^{2}}$ We do not claim that there is more gender discrimination in public accounting than other professional occupations. Madsen (2013) finds that women are not under-represented in public accounting when compared to other professional occupations, such as physicians, dentists, lawyers, clergy, financial managers, architects, engineers, chemists, veterinarians, and pilots.

clients that experience a change in the gender of the assigned partner (i.e., most prestigious clients are transferred from one male partner to another male partner).

In Belgium, compensation data are available for nearly half of the audit partners in our sample. To the extent that discrimination exists, we would expect female partners to receive lower compensation than male partners despite that female partners perform better in terms of generating higher fee premiums. We therefore compare the pay of male and female partners. The results provide strong evidence that female partners receive significantly lower pay than male partners. Specifically, the average compensation of male partners is 91,226 Euros higher than those of female partners, whose average compensation is 140,916 Euros. On its own, the evidence that female partners receive lower compensation does not necessarily mean that female partners are experiencing discrimination. Nevertheless, the results on the pay gap are more compelling when considered in tandem with our evidence that female partners perform better than male partners in terms of generating larger fee premiums for their audit firms. In other words, our findings suggest that female partners are paid less despite being more productive than male partners.

This study makes four contributions to the literature. First, we take a novel approach to testing gender discrimination by comparing the fees generated by male and female partners and their work assignments. This approach is different from the prior literature which tends to focus on the pay gap. We argue that it is crucial to compare the differences in productivity between men and women, not just their compensation. Our finding that female partners generate larger audit fees suggests that female partners have higher productivity (H1a). Importantly, this result is found to be stronger when female partners work in male-dominated offices (H1b). In addition, female partners tend to be assigned to less prestigious clients (H2a) despite that female partners

generate larger fee premiums (H1b). Moreover, the tendency for female partners to be assigned to less prestigious clients is particularly strong in male-dominated offices (H2b), where female partners are especially productive in terms of generating higher fees for their firms (H1b). Altogether, the combined findings for partner compensation, partner productivity, and client assignments are difficult to reconcile with other (non-discriminatory) theories relating to intrinsic differences between men and women.

Second, our study contributes to the literature by examining whether discrimination is attributable to employers (i.e., accounting firms) or their customers (i.e., clients). It could be argued that accounting firms do not want to discriminate against female partners but they do so because clients prefer male partners. Contrary to this argument, we show that clients pay fee premiums to female partners. Moreover, this result is more pronounced in offices with more male partners, which is where we would expect to observe the most discrimination. Together, these findings suggest that clients value female partners more than male partners, particularly when female partners work in male-dominated offices.

Third, prior studies rely on survey evidence to assess whether the public accounting profession suffers from gender discrimination (Stillabower 1985; Trapp et al. 1989; Anderson et al. 1994; Barker and Monks 1998; Komori 2008; Dalton et al. 2014). A limitation of the survey approach is that the reported perceptions of male and female accountants are very different. Men typically see little or no discrimination against women, whereas many women state that discrimination does exist (Stillabower 1985; Trapp et al. 1989). Instead of using survey data to test for discrimination, our study provides novel tests using large-sample archival data on audit fees, client assignments, and partner compensation. Overall, our results are consistent with the

perceptions of female accountants who report that there is significant discrimination in maledominated offices.

Fourth, our study offers an alternative explanation for the differences between male and female partners documented in the extant literature. Prior research finds that female partners are associated with higher audit fees and higher audit quality (Ittonen and Peni 2012; Ittonen et al. 2013; Hardies et al. 2015; Hardies et al. 2016). Researchers have inferred from this evidence that female accountants possess intrinsic advantages that make them superior to male accountants; e.g., women are more diligent, more conservative, and less overconfident than men (Reuben et al. 2012).³ Consistent with these studies, we find that female partners are associated with larger fee premiums.⁴ However, our explanation is different because we argue that the fee premiums are attributable to gender discrimination. The argument of prior studies that women are intrinsically better than men is inconsistent with our evidence that prestigious clients are usually assigned to male partners. Moreover, it is inconsistent with our evidence that male partners receive higher compensation. If women were inherently better, we would expect prestigious clients to be assigned to female partners and we would expect female partners to be paid more. However, we find the opposite. Moreover, our cross-sectional results for male-dominated offices lend additional support to our gender discrimination story.

A limitation of our study is that the analyses are confined to one service profession (accounting) in one country (Belgium). We believe that additional studies are necessary, particularly in countries with stronger legal protections for women (e.g., the U.S.). An important advantage of examining gender discrimination in the accounting profession (or other

³ In an untabulated analysis, we find statistically insignificant associations between accruals and the partner's gender.

⁴ Recent studies for the U.S. also find that female partners are associated with significant fee premiums (Burke et al. 2019; Lee et al. 2019).

professional services such as law firms) is that it is relatively straightforward to compare productivity differences between male and female partners using publicly available fee data. Section 2 discusses the prior literature on gender discrimination and develops the hypotheses. Section 3 describes the research design while Section 4 explains the data sources and provides descriptive statistics. Section 5 presents the results and Section 6 concludes.

2. Literature Review

2.1. What is Discrimination?

Gender discrimination can take various different forms. Taste-based discrimination arises when men and women are treated differently because of individual preferences. For example, a maledominated firm may indulge its preference for male employees by hiring fewer women or promoting women less often (Becker 1957). Another form of gender discrimination - statistical discrimination - occurs when the perceptions of male and female workers are based on gender stereotypes rather than individual characteristics (Phelps 1972). For example, an employer may choose not to promote a woman to a senior position because the employer believes that women are, on average, more interested in raising a family. The third type of discrimination - implicit discrimination - refers to the situation in which discrimination is unintentional (Bertrand et al. 2005). For example, a male boss may unconsciously treat a female subordinate differently than a male subordinate.

All three forms of discrimination (i.e., taste-based, statistical, implicit) are illegal in the workplace because they can result in unfair treatment. As discussed in the next two sections, prior studies attempt to test for gender discrimination by testing whether there are male-female disparities in compensation and job promotions. However, the results of these studies have

alternative explanations that are not necessarily based on discrimination. Our goal is to distinguish between these alternative explanations by examining how audit fees (a measure of partner productivity) and client assignments differ between male and female partners. More importantly, we examine how the findings vary across offices with relatively high (low) percentages of male partners. These cross-sectional office tests are important for our analysis because we expect that discrimination against women is more pronounced in offices that are more heavily dominated by male partners.

2.2. Male-Female Wage Disparities

There is considerable evidence that women are paid less than men and women are less likely to occupy the top positions at their firms (Matsa and Miller 2011; Gayle et al. 2012; Fernandez-Mateo and Fernandez 2016). The IMA 2012 Salary Survey reports that female accountants are paid only 78% of an average male accountant's salary (Schiffel et al. 2013). Schaefer and Zimmer (1995) show that the salary difference between male and female accountants is not fully explained by observable characteristics such as industry, location, education, and work experience. Expert testimony from an economist in the Kassman v. KPMG (2011) lawsuit alleges that the compensation of male senior audit managers is approximately \$53,000 higher than their female counterparts. In Belgium, the pay gap between men and women is 8% in the legal and accounting sector, which is slightly less than the 10% pay gap in other occupations (Institute for the Equality of Women and Men 2013).⁵

While lower pay is consistent with gender discrimination, other explanations are possible. For example, the salary differential might be due to hidden productivity differences between men and women (Oaxaca and Ransom 1994; Blau and Kahn 2000, 2017). An important limitation of

⁵ The Belgian Institute for the Equality of Women and Men reports the pay gaps across various economic sectors but it does not distinguish between law firms and accounting firms.

wage regressions is that the coefficient on the gender variable could capture unobserved differences in productivity rather than the effects of discrimination. To the extent that male workers are more productive than women in ways that are unobservable to the researcher, a wage regression will tend to ascribe the lower wages of women to discrimination rather than lower productivity. Therefore, in this study, we consider not only the compensation differences between male and female partners, but also the differences in their productivity. We test for productivity differences by examining whether male and female partners differ in their ability to generate fees for their audit firms.

2.3. Male-Female Disparities in Top Jobs

In the United States, 23% of partners are female whereas women have represented approximately 50% of all new hires by public accounting firms since the mid-1980s (AICPA 2019). Similar statistics have been reported in France (Dambrin and Lambert 2008), the Netherlands (de Accountant 2010), Sweden (Jonnergård et al. 2010), Canada (CPA Canada 2016), the United Kingdom (FRC 2017), Australia, and New Zealand (Whiting et al. 2015). However, these statistics do not necessarily imply that women are suffering from discrimination because alternative explanations are possible.

Niederle and Vesterlund (2007) suggest that women are less likely to occupy the top jobs because women shy away from highly competitive environments. The study argues that women are less competitive due to nurture (i.e., boys and girls are raised differently) and nature (i.e., men are more competitive due to the effects of human evolution). The lower incidence of women in the top jobs may also be attributable to women being more likely to take time out of the labor market due to motherhood or to raise a family (Lazear and Rosen 1990; Bertrand et al. 2010). Consistent with family responsibilities explaining the different career paths of men and women, Bertrand et al. (2010) show that female MBA graduates fall behind their male counterparts due to the conflicts between pursuing a career versus raising a family. Such conflicts are common in high-powered corporate jobs due to the long hours and heavy travel commitments that such positions demand (Goldin and Katz 2008; Bertrand et al. 2010). Long hours and heavy travel commitments are particularly important in public accounting, where employees are often required to work 60 to 80 hours per week during the busy season (Dalton et al. 1997; Gallhofer et al. 2011). Such commitments can put significant pressure on female accountants with primary family obligations. Consistent with motherhood being an important career impediment for female accountants, survey research finds that women are more likely than men to leave public accounting for family reasons (Dalton et al. 1997; Anderson-Gough et al. 2005).

In summary, a low incidence of female partners does not, in and of itself, provide compelling evidence of discrimination. Given the alternative explanations offered above, it is necessary to provide more rigorous evidence that women really do experience gender discrimination.

2.4. Anecdotal Cases of Alleged Gender Discrimination in Public Accounting

Employee lawsuits provide anecdotal examples of accounting firms being accused of discrimination (Price Waterhouse v. Hopkins 1989; Page v. PricewaterhouseCoopers 2004; Kassman v. KPMG 2011). Although the human resource practices of accounting firms have established gender diversity programs, female plaintiffs have alleged that such programs can hurt rather than help their promotion prospects (Anderson-Gough et al. 2005; Kornberger et al. 2010). For instance, in Page v. PricewaterhouseCoopers (2004), it was noted that the accounting firm had established a Mentoring Partnership Program that was compulsory for women but not men. The Program was established to help women move up the career ladder, but plaintiffs argued that

it simply added another level of scrutiny and opportunity for rejection that did not exist for their male colleagues.⁶

Many accounting firms offer reduced hours and flexible work schedules to help working mothers (Gallhofer et al. 2011). However, these arrangements can have unintended consequences because, when a woman takes a reduced or flexible schedule, it can be perceived by her male counterparts as signaling the woman's lack of commitment to the firm.⁷ Such perceptions are apparent in the lawsuit of Kassman v. KPMG (2011), where a male partner allegedly told a female plaintiff that "*part time women work exactly their x hours and not a minute over*", "*you know you can't make them work anymore*", and "*you know they're not going to get anywhere.*" Accountants on the partner track need to spend a lot of time socializing and networking with clients and fellow partners (Kumra and Vinnicombe 2008). However, female accountants can find themselves excluded when social events are designed to suit the tastes of men rather than women (Anderson-Gough et al. 2005; Kumra and Vinnicombe 2008). For example, in Page v. PricewaterhouseCoopers (2004), it was alleged that the firm had:

"excluded the [female] Plaintiff from social events, golf outings, and other network opportunities designed to allow employees to develop and enhance relationships with clients and partners, while males were routinely allowed and encouraged to participate

⁶ Class action lawsuits are not permitted in Belgium. Therefore, we discuss the evidence from class action lawsuits in the United States instead.

⁷ Plaintiffs in Kassman v. KPMG (2011) claimed that: "Although KPMG touts its flexible work schedule as proving it supports working mothers, the reality is that working mothers are often forced to take a so-called "reduced" schedule for less pay – but are still expected to shoulder the same responsibilities as their full-time counterparts. Many of KPMG's female employees feel pressured to move to a flexible plan after having children due to the Company's stereotype that working mothers with young children are less effective employees and less committed to their careers. These mothers do not accept reduced pay by choice, but to avoid heightened scrutiny and termination. [...]KPMG discriminates against its female employees do not have to overcome." This allegation was supported by several female plaintiffs having their employments terminated by KPMG on their first day back from maternity leave. Female accountants can find ways to signal their long-term commitment despite motherhood. In Kassman v. KPMG (2011) it was further alleged that the firm had "frequently touted a Partner in the Atlanta office, Tammy Hunter, as a role model for other working mothers because she called the Company to discuss work matters on her way to the delivery room to give birth to her child. The Company apparently viewed this as a level of dedication to which other working mothers should aspire."

in such events. For example, even on accounts where Plaintiff held major responsibility, her male co-workers were invited to poker parties, Las Vegas trips, and golf outings, but Plaintiff was excluded. On the rare occasion when Plaintiff was invited, the invitation was extended in a way that was not sincere. For example, Plaintiff was told "fifteen guys are going to Las Vegas – do you really want to go?" Additionally, PWC events were tailored toward male preferences such as golf and sporting events."⁸

2.5. Hypotheses Development For Audit Fees: H1a and H1b

In this study, we compare the productivity of male and female partners using the fees that partners generate for their audit firms. In most occupational settings, productivity differences between men and women are difficult to measure. Public accounting is an advantageous setting because the fees generated by partners are publicly disclosed for every audit client.

There is anecdotal evidence that accounting firms sometimes discriminate against women by promoting men who exhibit inferior productivity. For instance, in Price Waterhouse v. Hopkins (1989), the Supreme Court concluded that the firm had denied the only woman considered for partnership even though she had generated higher fees than the other 88 male candidates. In Page v. PricewaterhouseCoopers (2004), it was alleged that the accounting firm had favored two male candidates over the female plaintiff, despite the men working fewer hours, generating lower fees, and receiving inferior performance evaluations compared to the female candidate.

We argue that gender discrimination is likely to cause differences in the observed performance of male and female partners. To see why, consider Figure 1, where men and women are assumed to have the same ability distributions but the threshold for promotion to the partner level is set higher for women. Figure 1 shows that the women who are promoted to partner

⁸ Male colleagues had described the female plaintiff as "overbearing, macho and abrasive and said she would have a better chance of making partner if she would wear makeup and jewelry, and walk, talk, and dress more femininely." The Supreme Court concluded that the firm had "permitted negative sexually stereotyped comments to influence partner selection."

perform better than the men who are promoted to partner because women have to meet a higher performance threshold in order to be promoted.

[INSERT FIGURE 1 HERE]

Given the higher performance threshold for female partners relative to male partners, we hypothesize that female partners generate higher audit fees than male partners (H1a).

HYPOTHESIS 1a. Female partners are associated with higher fees than male partners.

A limitation of H1a is that female partners may generate higher fees for reasons unrelated to discrimination. For example, women may have intrinsic characteristics that make them better accountants than males: women may be more risk-averse, more conscientious, and more conservative than men (Mueller and Plug 2006; Croson and Gneezy 2009; Reuben et al. 2012; Hardies et al. 2015). To assess whether the fee differential between male and female partners is really due to gender discrimination, we examine whether the differential is larger in offices with a higher proportion of male partners.

Prior research suggests that discrimination is related to the gender composition of the workplace. For example, Hultin and Szulkin (1999) find that women in Sweden earn less than men if they work in a company with more male managers. Gorman (2005) shows that male recruiters are less likely than female recruiters to hire female job candidates in American law firms. Similarly, Kurtulus and Tomaskovic-Devey (2012) find that an increase in the share of male top managers is associated with fewer female middle managers being hired. In public accounting, Dalton et al. (2014) show that female accountants perceive more discrimination in offices that have a higher percentage of male partners. Therefore, if the fee differential between male and female partners is attributable to gender discrimination, we would expect the differential to be more pronounced in offices with relatively more male partners.

HYPOTHESIS 1b. The positive association between audit fees and female partners is significantly larger in offices with a higher percentage of male partners.

2.6. Hypotheses Development For Client Assignments: H2a and H2b

A partner can have a more rewarding career when the partner works on more prestigious audit engagements (Knechel et al. 2013). Therefore, if female partners encounter discrimination, we would expect prestigious clients to be assigned to male partners with non-prestigious clients being assigned to female partners. There is anecdotal evidence consistent with this idea. In Page v. PricewaterhouseCoopers (2004), the female plaintiff alleged that the firm had "enhanced a male candidate's potential for partnership by assigning him to one of the [female] plaintiff's most lucrative clients. This particular client was one of the largest in the Chicago office and was brought to the firm by the [female] plaintiff." To the extent that partner-client assignments serve to discriminate against female partners, we would expect prestigious clients to be assigned to male rather than female partners. This leads to our next hypothesis.

HYPOTHESIS 2a. Prestigious clients are less likely to be assigned to female partners than male partners.

A limitation of H2a is that male partners may be assigned to prestigious clients for other reasons. For example, prestigious clients may require the attention of a full-time partner working long uninterrupted hours (Kornberger et al. 2010). To the extent that female partners are more likely to face family disruptions, this could mean that female partners are less suitable for working on the accounts of prestigious clients. Another possibility is that prestigious companies prefer to be audited by male rather than female partners (Lee et al. 2019), or female partners may prefer to work with less demanding clients.

To determine whether our H2a results are really attributable to gender discrimination, we examine whether the difference in client assignment patterns is more pronounced in the offices that have a higher percentage of male partners. Prior research suggests that women are likely to face more discrimination when they work in male-dominated environments (Hultin and Szulkin 1999; Gorman 2005; Cohen and Huffman 2007; Kurtulus and Tomaskovic-Devey 2012; Dalton et al. 2014). It follows that, if the client assignment patterns predicted in H2a are attributable to gender discrimination, those patterns would be more evident in the offices that have relatively more male partners. This leads to our final hypothesis:

HYPOTHESIS 2b. The client assignment patterns predicted in H2a are significantly stronger in offices that have a higher percentage of male partners.

2.7. Alternative Arguments

Forces other than gender discrimination could potentially drive different outcomes for male and female partners. We discuss these other forces and explain how they generate predictions that differ from one or more of our predictions in H1a, H2a, H1b, and H2b.

1) Client preferences. One possibility is that clients prefer to be audited by male rather than female partners. If so, we would expect clients to pay higher fees to male partners. This would be opposite to our prediction in H1a that clients pay higher fees to female partners.⁹ Another possibility is that prestigious clients have a stronger preference for male partners, or that female partners prefer to be assigned to less prestigious clients. This would be consistent with our prediction in H2a that prestigious clients are more likely to be assigned to male partners. However, it would not generate the prediction in H2b that this pattern is stronger in the male-

⁹ Contrary to the client preferences hypothesis, there is survey evidence that female and male partners are equally acceptable to audit clients (Trapp et al. 1989).

dominated offices. Therefore, the client preferences hypothesis makes predictions that are inconsistent with our predictions in H1a and H2b.

2) Superior intrinsic ability of male accountants. Another argument is that male partners are intrinsically better than female partners because men are more competitive, or have fewer family commitments, or are less likely to take a career break (Niederle and Vesterlund 2007; Sutter and Glätzle-Rützler 2015; Saccardo et al. 2018). If men really perform better than women, we would expect male partners to be more productive. This would be opposite to our prediction in H1a that female partners generate larger fees for their firms.

3) Superior intrinsic ability of female accountants. An opposite argument is that female partners are intrinsically better than male partners because women are more diligent, more conservative, and less overconfident (Ittonen et al. 2013; Hardies et al. 2015). If female partners are truly superior, we would expect prestigious clients to be assigned to female partners, which is opposite to our prediction in H2a. Moreover, if female partners are truly superior, we would not expect them to be paid less than male partners.

4) Differences in labor supply. It is plausible that an office will have more male partners when the office is located in an area that has a more plentiful supply of talented males. According to this argument, the male partners in male-dominated offices would be more talented and therefore associated with higher fees. However, this is opposite to our prediction in H1b that the lower fees generated by male partners are more pronounced in male-dominated offices.

5) Self-selection. The women who choose to work in male-dominated offices may have intrinsic characteristics that are more similar to those of men (Nekby et al. 2008). According to this self-selection argument, the fee differentials between men and women would be smaller in male-dominated offices because the men and women in male-dominated offices would be more

similar. This is opposite to our prediction in H1b that the fee differences between male and female partners are larger in male-dominated offices. Similarly, the association between prestigious clients and gender would be smaller in male-dominated offices if the men and women in these offices are more similar. Again, this is opposite to our prediction in H2b that the tendency for prestigious clients to be assigned to male partners is stronger in male-dominated offices.

6) Affirmative action. An office may follow a policy of affirmative action by setting a lower performance threshold for female partners (i.e., the opposite to Figure 1). A lower performance threshold would result in less productive women being promoted to the partner level, which would be opposite to our prediction in H1a that female partners generate higher fees. The affirmative action story would also result in male-dominated offices having smaller fee differentials between male and female partners because affirmative action should make it less likely that offices are dominated by less productive men. This would be opposite to our prediction in H1b that the fee differential between male and female partners is larger in male-dominated offices.

[INSERT TABLE 1 HERE]

In summary, the above six arguments - client preferences, greater male productivity, greater female productivity, differences in labor supply, self-selection, and affirmative action - generate predictions that are different from our hypotheses (see Table 1 for a summary). Therefore, if our results collectively support H1a, H1b, H2a, and H2b, the findings are unlikely to be explained by any of these alternative arguments, but are instead more consistent with gender discrimination.

3. Research Design

We test H1a by estimating the following model of audit fees.

$$LAF = \alpha_0 + \alpha_1 FEMALE + CONTROLS + u$$
(1a)

The dependent variable is the log of audit fees (*LAF*). The *FEMALE* variable equals one if the audit partner is a woman, and zero if a man. Under H1a, we expect female partners generate higher audit fees. Therefore, we predict a positive coefficient on *FEMALE* (i.e., $\alpha_1 > 0$).

Following prior studies on the determinants of audit fees (Hay et al. 2006), our control variables include client size, an indicator for whether the company is publicly traded, and various proxies for the company's financial health (e.g., profitability, leverage, and liquidity). We control for the size of the audit firm, whether the audit firm is an industry specialist, whether the partner is an industry specialist, whether the partner specializes in serving public companies, the length of the partner's career, the number of clients served by the partner, the partner's audit opinion, and whether the partner works in a French-speaking area of Belgium. The variables are defined in the Appendix. We also control for city fixed effects, year fixed effects, and industry fixed effects.

To test H1b, we measure the percentage of male partners in each office (*OMALE*%). A higher value of *OMALE*% corresponds to an office that has relatively more male partners. We augment eq. (1a) by including *OMALE*% and its interaction with the gender of the audit engagement partner (*FEMALE*).

$$LAF = \beta_0 + \beta_1 FEMALE + \beta_2 OMALE\% + \beta_3 FEMALE \times OMALE\% + CONTROLS + u \quad (1b)$$

Under H1b, the fee premiums of female partners are significantly larger in offices with more male partners. We therefore predict a positive coefficient on the *FEMALE* × *OMALE*% interaction variable in eq. (1b) (i.e., $\beta_3 > 0$).¹⁰

We test H2a and H2b by estimating models that explain the assignments of prestigious and non-prestigious clients to male and female partners. Because auditor-client assignments are determined through a matching process, it is unclear whether the dependent variable should capture the gender of the partner or the prestige of the client. On one hand, if the partner is chosen for a specific client, the dependent variable should be the partner's gender (*FEMALE*). On the other hand, if the client is chosen for a specific partner, the dependent variable should be *PRESTIGE*. To address this issue, we test H2a and H2b using both dependent variables as alternatives:

$$FEMALE = \delta_0 + \delta_1 PRESTIGE + CONTROLS + v$$
(2a)

$$PRESTIGE = \theta_0 + \theta_1 FEMALE + CONTROLS + v$$
(2b)

The *PRESTIGE* variable equals one if the client is publicly traded, a subsidiary of a publicly traded company, or the client is in the top 10% of the office's largest clients (measured using total assets), and equals zero otherwise.¹¹ Under H2a, we expect prestigious clients are less likely to be assigned to female partners. We therefore predict a negative coefficient for *PRESTIGE* in eq. (2a) (i.e., $\delta_1 < 0$) and a negative coefficient for *FEMALE* in eq. (2b) (i.e., $\theta_1 < 0$).

Eq. (2a) controls for client characteristics that could affect whether the client is assigned to a male or female partner. These include the company's financial health (e.g., profitability,

¹⁰ The β_1 coefficient in (1b) captures the effect of *FEMALE* in offices with zero male partners (i.e., when *OMALE*% = 0). We do not make a prediction for the sign of β_1 because, in our sample, there are very few female-only offices. Instead, we interpret the economic magnitude of the *FEMALE* coefficient as we vary the value of *OMALE*% from the 25th percentile value to the 75th percentile value.

¹¹ Although the 10% threshold is somewhat arbitrary, we obtain similar inferences using alternative thresholds (e.g., 20%) and/or using sales rather than assets to measure client prestige.

leverage, and liquidity), the provision of non-audit services, and the size of the accounting firm. Eq. (2b) controls for partner characteristics that could determine whether the partner is assigned to a prestigious client. These include whether the partner is an industry specialist, whether the partner specializes in serving public companies, the length of the partner's career, the partner's number of clients, and whether the partner works in a French-speaking area of Belgium. In each equation, we also control for city fixed effects, year fixed effects, and industry fixed effects.

To test H2b, we augment eqs. (2a) and (2b) by adding interactions with the office gender variable (*OMALE*%).

 $FEMALE = \lambda_0 + \delta_1 PRESTIGE + \delta_2 OMALE\% + \delta_3 PRESTIGE \times OMALE\% + CONTROLS + v$ (2c)

 $PRESTIGE = \theta_0 + \theta_1 FEMALE + \theta_2 OMALE\% + \theta_3 FEMALE \times OMALE\% + CONTROLS + v$ (2d)

Under H2b, the tendency for prestigious clients to be assigned to male partners is significantly stronger in offices with a higher percentage of male partners. Therefore, we predict a negative coefficient on *PRESTIGE* × *OMALE*% in eq. (2c) (i.e., $\delta_3 < 0$) and a negative coefficient on *FEMALE* × *OMALE*% in eq. (2d) (i.e., $\theta_3 < 0$).¹²

4. Institutional Setting, Sample, and Descriptive Statistics

4.1. Institutional setting

Belgium's accounting and auditing regulations are very similar to other EU countries (Vanstraelen and Willekens 2008; Hardies et al. 2018). All public and 'large' private companies

¹² Again, we do not make predictions for δ_1 or θ_1 because there are very few female-only offices in our sample. Instead, we examine how the assignments of prestigious clients to male and female partners change as the value of *OMALE*% increases from the 25th percentile to the 75th percentile.

are legally required to publish full audited financial statements.¹³ The use of International Financial Reporting Standards (IFRS) is obligatory for the consolidated accounts of publicly listed companies, and optional for the consolidated accounts of all other companies. Audit firm appointments are made for three years, and can be renewed for additional three-year periods (Vanstraelen and Willekens 2008). Partner rotation is required after six years for public interest entities (Directive 2006/43/EU), with a two-year cooling-off period before the partner is permitted to rotate back onto the engagement.

In 2002, Belgium created an Institute for Equal Opportunities between Women and Men (IGVM) in order to help address gender discrimination. A law against gender discrimination was adopted in 2007 and amended in 2018. However, Belgium is a low litigation environment (Vanstraelen and Willekens 2008; Hardies et al. 2018) and lawsuits alleging discrimination are rare. Therefore, firms in Belgium have weak litigation-based incentives to not discriminate against female employees. The influx of women in accounting has been relatively slow in Belgium. In the 1990s the number of female CPAs in Belgium was barely above 10%, so the Belgian accounting profession has been male-dominated until relatively recently. Currently, the proportion of female CPAs is lower in Belgium than the United States (approx. 25% vs. 50%). Furthermore, only 15% of Belgian partners are women whereas 23% of partners are women in the United States (AICPA 2019).

4.2. Sample

¹³ During our sample period (2008–2014), companies were considered 'large' if they exceeded at least two of the following size thresholds: (1) revenues (excluding VAT) > 7,300,000 euros; (2) total assets > 3,650,000 euros; and (3) number of employees > 50. These criteria had to be considered on a consolidated basis if the company belonged to a group or was a holding or listed company. Additionally, companies were always considered 'large' if they were listed or had more than 100 employees.

Our sample starts with every audited company over the period 2008-2014. The data are from the *Bel-First* database, *Graydon Belgium*, and the Belgian National Bank, which maintains financial statements for public and private companies. Partner names are from *Graydon Belgium* and manually verified against the names shown in audit reports to ensure their accuracy. For partners whose gender could not be determined from the first name, we checked several sources to identify the partner's gender (e.g., LinkedIn and the Institute of Registered Auditors).

As shown in Table 2, our sample starts with 127,076 client-year observations in 2008-2014. We drop 4,641 observations with joint audit firms and we drop 11,288 sole proprietorship observations because a sole proprietor is unlikely to face discrimination from her own firm. We lose 5,953 observations due to missing data on audit fees. We drop 10,128 observations for financial institutions and public administrative institutions because their accounting and auditing requirements are different. We lose 184 observations due to missing information for the control variables. This leaves a final sample of 94,882 company-year observations. The final sample has 656 unique partners of which 100 (15.0%) are women.¹⁴ There are 112 unique accounting firms and a total of 199 offices. On average each office has 85 clients, of which 11 are classified as prestigious, and each office has an average of 3 audit partners.

[INSERT TABLE 2 HERE]

4.3. Descriptive statistics

Panel A of Table 3 reports descriptive statistics for the clients of male and female partners. Consistent with H1a, we find that female partners are associated with significantly higher audit fees (*LAF*) (t-stat. = 6.39) despite that their clients are significantly smaller (t-stat. = -3.84).

¹⁴ The Hardies et al. (2015) study has 692 unique partners of which 93 are women. There are fewer partners in our study because we exclude the sole proprietors.

Consistent with H1a, these findings suggest that female partners are associated with higher audit fees. Consistent with H2a, we find that only 10.6% of female partner clients are prestigious whereas 13.3% of male partner clients are prestigious.

[INSERT TABLE 3 HERE]

Panel B of Table 3 reports descriptive statistics after partitioning the sample into maledominated offices ($0.750 \le OMALE\% \le 1.000$; N = 79,974) and offices with a more diverse gender balance ($0.000 \le OMALE\% < 0.750$; N = 14,908). We find that audit fees are on average higher in the male-dominated offices. However, this could also reflect a clientele effect because the clients in male-majority offices are more prestigious and tend to be larger (in terms of assets and sales). Table 4 provides a correlation matrix for the variables. All of the correlations are below 0.5 and we find no evidence of multicollinearity problems in our regressions based on the variance-inflation-factors.

[INSERT TABLES 3 & 4 HERE]

5. Results

5.1. Audit Fee Regressions (H1a and H1b)

Table 5 begins by reporting regression results for the models of audit fees. In models 1 and 2, the coefficients on *FEMALE* are positive and significant (t-stats. = 2.96, 2.41). In economic terms, these coefficients translate into fee premiums of 3.98% and 3.25% for female partners. This supports the prediction in H1a that female partners generate higher audit fees than male partners.

[INSERT TABLE 5 HERE]

Models 3 and 4 test H1b by including the gender composition of the office (*OMALE*%) and its interaction with the gender of the audit engagement partner (*FEMALE*). The interaction variable (*FEMALE* × *OMALE*%) has significant positive coefficients (t-stats. = 8.71, 5.95). This

strongly supports the prediction in H1b that the fee premiums to female partners are significantly larger in male-dominated offices. To assess the economic significance of these results, we examine how the fee premium to female partners varies between offices with the 25th and 75th percentile values of *OMALE*%. Over this range and using the coefficient estimates from model 4, we calculate that the average fee premiums to female partners increase from 3.3% to 11.1%. Therefore, the office gender variable (*OMALE*%) has an economically significant effect on the fee differential between male and female partners. We conclude that the fee premiums to female partners are significantly larger in male-dominated offices.

Results for the control variables are consistent with prior research (Hay et al. 2006). The coefficients on *LTA*, *LNSALES*, and *LIST* show that large companies and publicly traded companies pay higher audit fees. Fees are larger when companies are more highly leveraged, less profitable, have more current assets, lower liquidity, and more non-audit services. In addition, audit fees are higher when the audit partner is a public company specialist, the partner became a CPA more recently, the partner is from a Big 4 firm, and the partner has fewer clients.

5.2. Aggregate Abnormal Audit Fees For Each Partner-Year (H1a and H1b)

It could be argued that our analyses should be conducted at the partner-year level rather than the audit engagement level. We examine the implications of moving the analysis to the partner-year level in the following three steps. First, we estimate abnormal audit fees for each engagement using the residuals from the audit fee model in Col. 2 of Table 5 (but without the *FEMALE* variable). Next, we cumulate the values of abnormal audit fees across all engagements of a given partner-year (3,468 partner-years in total). Finally, we test whether aggregate abnormal audit fees at the partner-year level are larger for female partners than male partners (H1a); and, we test whether this positive association is driven by male-dominated offices (H1b).

The univariate results are shown in panel A of Table 6. In the male-dominated offices $(0.750 \le OMALE\% \le 1.000)$ we find that aggregate abnormal audit fees are significantly larger for female partners than male partners (t-stat. = 4.38).¹⁵ In contrast, aggregate abnormal audit fees are not significantly different between female and male partners in offices with a more diverse gender balance $(0.000 \le OMALE\% < 0.750)$. The regression results are reported in panel B of Table 6.¹⁶ Consistent with H1a, we find that aggregate abnormal audit fees are significantly larger for female partners (t-stats. = 3.17, 3.10). Consistent with H1b, the positive association between aggregate abnormal audit fees and female partners is significantly larger in offices that have a higher proportion of male partners. We conclude that our findings for H1a and H1b continue to hold when we conduct the analyses at the individual partner level rather than the audit engagement level.

[INSERT TABLE 6 HERE]

5.3. The Annual Compensation of Male and Female Partners

In Belgium, close to half of the audit partners are compensated by their audit firms through their own individual management companies. These management companies are required to file annual financial statements with the National Bank of Belgium, which means we are able to obtain information on individual partner incomes. Following Vandenhaute et al. (2020), we use the incomes of audit partners' individual management companies as a proxy for their compensation. We obtain the compensation data for 1,343 of the 3,468 partner-year observations

¹⁵ In an untabulated test, we drop the offices with zero females and we continue to find that aggregate abnormal audit fees are significantly larger for female partners in the male-dominated offices (i.e., where $(0.750 \le OMALE\% \le 1.000)$).

¹⁶ In the Table 6 regressions, we control for characteristics of an audit partner's client portfolio that could affect their ability to generate abnormal audit fees. Specifically, we control for office size (*OFFICE_SIZE*), the mean of the natural log of total assets for partner *i*'s clients in year *t* (*MEAN_LTA*), the mean ratio of liabilities to assets for partner *i*'s clients in year *t* (*MEAN_LEV*), and the mean return on assets for partner *i*'s clients in year *t* (*MEAN_ROA*).

in our sample. This sample comprises 178 female partner-years and 1,165 male partner-years. Our expectation is that female partners are paid less than male partners despite that female partners generate higher fees for their audit firms.

Panel A of Table 7 reports the univariate differences in male and female partner compensation. The average compensation of male partners is 232,142 Euros whereas the average compensation of female partners is only 140,916 Euros and the difference is highly significant (t-stat. = -4.93).¹⁷ The difference in pay remains significant when we use a log transformation for compensation (t-stat. = -4.89). Panel B of Table 7 reports the results of regression models that control for the same partner characteristics as Table 6. Models 1 and 2 confirm that female partners earn significantly less than male partners (t-stats. = -2.81, -3.50). On its own, the fact that female partners are paid less does not necessarily imply gender discrimination. However, the compensation evidence is more compelling when considered in conjunction with our evidence that female partners generate significantly larger fees for their audit firms. In other words, female partners are paid less despite being more productive in terms of generating higher fees.

[INSERT TABLE 7 HERE]

In Tables 5 and 6, we found that the difference in male-female productivity is especially pronounced in male-dominated offices (H1b). To the extent that the female partners in male-dominated offices are rewarded for their greater productivity, we would expect the female pay discount to be significantly smaller in male-dominated offices. On the other hand, it is plausible that the female pay discount might be larger in the male-dominated offices given that these are

¹⁷ In an untabulated test we find that partner compensation is generally higher at the Big 4 firms than at the non-Big 4 firms but female partners in Big 4 and non-Big 4 firms tend to earn lower compensation than their male counterparts. In Big 4 firms, the average compensation of male partners is 384,989 Euros whereas the average compensation of female partners is 213,397 Euros and the difference is highly significant (t-stat. = -4.98). In non-Big 4 firms, the average compensation of male partners is 111,881 Euros whereas the average compensation of female partners is 96,109 Euros and this difference is also significant (t-stat. = -2.08).

the offices where discrimination is most apparent. Accordingly, we do not make a signed prediction for the *FEMALE* × *OMALE*% interaction variable in the compensation regressions. In models 3 and 4, we find significant positive coefficients on the *FEMALE* × *OMALE*% interaction (t-stats. = 1.72, 3.59). Therefore, the female partners in male-dominated offices are paid more than the female partners in other offices.

5.4. Client Assignments (H2a and H2b)

Table 8 reports tests of H2a and H2b using the logit models of client assignments. The dependent variable is *FEMALE* in panel A and *PRESTIGE* in panel B. In model 1 (panel A) the coefficient on *PRESTIGE* is negative and significant (z-stat. = -3.14). In model 1 (panel B) the coefficient on *FEMALE* is negative and significant (z-stat. = -3.69). These results strongly support the prediction in H2a that prestigious clients are less likely to be assigned to female partners. We test H2b by adding the interactions with the office variable (*OMALE*%). In model 2 (panel A) the coefficient on *PRESTIGE* × *OMALE*% is negative and significant (z-stat. = -2.44). In model 2 (panel A) the coefficient on *FEMALE* × *OMALE*% is negative and significant (z-stat. = -2.69). These findings support the prediction in H2b that the tendency for firms to assign non-prestigious clients to female partners is stronger in the offices that have relatively more male partners.

[INSERT TABLE 8 HERE]

We assess the economic significance of these H2b results by calculating how the probability of a prestigious client being assigned to a male partner increases as the value of *OMALE*% increases from the 25th percentile value (79%) to the 75th percentile (100%). Over this range and using the coefficient estimates in model 2 (panel A), the predicted probability of a prestigious client being assigned to a male partner is found to increase from 96.0% to 99.3%. We

conclude that the influence of *OMALE*% on the assignments of prestigious clients to female and male partners is economically significant as well as statistically significant.

5.5. Female Partners Attracting New Clients and Losing Prestigious Clients

We examine whether male and female partners differ in their ability to attract new clients to the audit firm. For this analysis, we assume the partner who brings the client to the audit firm is the partner who works with the client during the first year of the engagement. (We cannot directly observe which partner is actually responsible for bringing clients to the audit firm.) Out of 72,762 firm-year observations for which there is prior year information, we observe 5,336 audit firm changes (i.e., new clients for the audit firm). Of these, we find that 654 (12.3%) are audited by female partners. This figure is higher than the 10.4% of retained clients audited by female partners, which suggests that female partners spend relatively more time attracting new clients to the audit firm rather than auditing existing clients.

Next, we examine instances of audit partner rotations to see which types of clients are rotated between female and male partners. After dropping the 5,336 partner changes that occur due to a change of audit firm, we find 4,911 audit partner rotations involving the same audit firm. Next, we split the 4,911 partner rotations between those involving prestigious and non-prestigious clients. Out of 815 prestigious clients that switch audit partners, we find 63 cases in which the initial audit partner was female. In all 63 cases, we find the prestigious client is subsequently rotated to a male partner. Out of 752 prestigious clients that initially had a male partner, we find that 87% rotate to another male partner while only 13% rotate to a female partner. In contrast, we find that female partners are more likely to be rotated onto the engagements of non-prestigious clients. There are 4,096 non-prestigious clients that switch audit partner was male, 16% switch to a female

partner and out of 345 partner changes where the initial audit partner was female, 13% switch to another female partner.

Overall, these findings suggest that female partners are more likely to be rotated onto the engagements of non-prestigious clients (13% and 16%) whereas female partners are less likely to be rotated onto the engagements of prestigious clients (0% and 13%). This is consistent with recent evidence in Dodgson et al. (2020) that client assignments during partner rotations are non-random. Nevertheless, we are cautious with these findings given the relatively small number of rotations that involve both female partners and prestigious clients.

5.6. Propensity Score Matching

A potential concern is that the clients of male partners may be different from the clients of female partners. Another concern is that there could be systematic differences between male and female partners (e.g., industry expertise, number of clients, years of experience). Although we control for client characteristics, partner characteristics, and audit firm characteristics in the regressions, there could still be a concern that the assumed functional form for the effects of the control variables may be incorrect.

To address these concerns, we match the clients of female partners to the clients of male partners based on clients' and auditors' observable characteristics. Matching allows us to avoid making any assumptions about the appropriate functional forms (Shipman et al. 2017). We first estimate a client assignment model for the full sample in which the dependent variable is *FEMALE*. The model is estimated using both client characteristics (*LTA*, *LNSALES*, *LIST*, *LEV*, *ROA*, *CATA*, *QUICK*, *IRISK*, *LOSS*, *PROBF*, *LNAS*, *BUSY*, *OPINION*), audit firm characteristics (*BIG4*, *SPECF*), and audit partner characteristics (*SPECAP*, *PUBSPEC*, *LNCAREER*, and *LANG*). We use one-to-one-matching without replacement and with a caliper of 0.2.¹⁸ The propensity-matched sample comprises 19,814 observations. Next, we re-estimate the audit fee regressions using the matched sample. We find that the matched sample results are very similar to our tabulated analyses.

5.7. Offices with a Mix of both Male and Female Partners (0% < OMALE% < 100%)

In our sample, 42,688 observations (44.99%) belong to all-male offices (OMALE% = 100%) and 952 observations (1.00%) belong to all-female offices (OMALE% = 0%). It is not obvious that these observations should be included in our tests. On one hand, it can be argued that they should be included because the male partners in male-only offices may have been promoted to the partner level despite being less productive than women. We would expect this to show up in abnormally large fee discounts for the male partners in male-only offices. On the other hand, client assignments may be decided at the office-level rather than the firm-level, in which case there would be no opportunity for male-only offices to discriminate against female partners when making client assignment decisions.

To assess whether our results are robust, we re-estimate the fee and client assignment models after restricting the sample to offices with a mix of both male and female partners (N = 51,242). In the audit fee model, we continue to find a significant positive coefficient on *FEMALE* × *OMALE*% (t-stat. = 2.97). This supports our conclusion that female partners generate larger fees when they work in male-dominated offices. In the client assignment models, we continue to find significant negative coefficients on *PRESTIGE* × *OMALE*% and *FEMALE* × *OMALE*% (z-stats. = -2.89, -3.00). Therefore, the tendency for non-prestigious clients to be assigned to female partners is significantly stronger in male-dominated offices.

¹⁸ We find similar results when we match with replacement or when we use alternative calipers.

5.8. Fees from Non-audit Services

Our analysis has focused on audit fees because non-audit fees are usually negotiated by nonaudit partners rather than audit partners. Nevertheless, audit firms can receive non-audit fee income when audit partners refer their clients to advisory partners. We therefore integrate the fees from non-audit services into our analysis of partner productivity. We calculate a new fee variable that equals the client's total fee for audit and non-audit services. We then re-estimate the fee models after replacing the dependent variable (log of audit fees) with the log of total fees. We find very similar results in this alternative specification. The coefficient on *FEMALE* × *OMALE*% is positive and significant (t-stat. = 3.32), indicating that the fee premiums to female partners are significantly larger in the male-dominated offices.

6. Conclusion

Prior research reports that women are paid less than men and women are less likely to be promoted to the top jobs. Although this evidence is consistent with discrimination, there are other explanations which emphasize the intrinsic differences between men and women, including differences in their performance, preferences, and career interruptions. We address these identification issues by taking a new approach to testing for gender discrimination using the public accounting industry as our setting.

If accounting firms discriminate against women in partner promotion decisions, we would expect female partners to perform better than male partners. We test this by comparing the fee differential between male and female partners. We find that female partners generate significantly higher fees than male partners. Importantly, this result is found to be stronger in offices with a higher proportion of men, which is where we would expect to find the most

discrimination. Further, we find that prestigious clients are less likely to be assigned to female partners and that this result is stronger in offices with a higher proportion of males. Overall, we argue that these results are consistent with gender discrimination.

We emphasize that we cannot directly observe gender discrimination and our most important tests depend on an assumption that discrimination (if it exists) is more likely to be present in offices with a higher representation of male partners. Collectively, we argue that it is difficult to attribute our results to other explanations, such as client preferences, differences in the intrinsic abilities of male and female accountants, differences in the supply of male and female accountants, self-selection of employee types to different types of offices, and affirmative action policies. Nevertheless, we cannot rule out the possibility that other unexplored gender differences might account for our results.

A limitation of our analysis is that it relies on data from just one country (Belgium). We acknowledge that our results may not generalize to other countries, where it is easier for women to sue their employers (e.g., the U.S.). Moreover, there is circumstantial evidence that discrimination may be more severe in Belgium than in other developed countries. In Belgium, only 15% of partners are women whereas in the United States 23% of partners are women (AICPA 2019). Moreover, only 11.0% of board positions are occupied by women in Belgium compared to 14.0% in other European countries. Given these statistics, it remains an open question whether our findings are generalizable to other countries.

References

- Anderson, J. C., E. N. Johnson, and P. M. J. Reckers. 1994. Perceived effects of gender, family structure, and physical appearance on career progression in public accounting: A research note. Accounting, Organizations and Society 19: 483-91.
- Anderson-Gough, F., C. Grey, and K. Robson. 2005. Helping them to forget: The organizational embedding of gender relations in public audit firms. *Accounting, Organizations and Society* 30: 469-90.
- AICPA (American Institute of Certified Public Accountants). 2019. Trends in the supply of accounting graduates and the demand for public accounting recruits. New York, NY: AICPA.
- Barker, P. C., and K. Monks. 1998. Irish women accountants and career progression: A research note. *Accounting, Organizations and Society* 23: 813-23.
- Becker, G. S. 1957. The Economics of Discrimination. University of Chicago Press, Chicago.
- Bertrand, M., D. Chugh, and S. Mullainathan. 2005. Implicit discrimination. *American Economic Review* 95: 94-98.
- Bertrand, M., C. Goldin, and L. Katz. 2010. Dynamics of the gender gap for young professionals in the financial and corporate sectors. *American Economic Journal: Applied Economics* 2: 225-55.
- Blau, F. D., and L. M. Kahn. 2000. Gender differences in pay. *Journal of Economic Perspectives* 14: 75-99.
- Blau, F. D., and L. M. Kahn. 2017. The gender wage gap: Extent, trends, and explanations. *Journal of Economic Literature* 55: 789-865.
- Burke, J., R. Hoitash, and U. Hoitash. 2019. Audit partner identification and characteristics: Evidence from U.S. Form AP filings. *Auditing: A Journal of Practice & Theory* 38: 71-94.
- Cohen, P. N., and M. L. Huffman. 2007. Working for the woman? Female managers and the gender wage gap. *American Sociological Review* 72: 681-704.
- CPA Canada. 2016. Annual Report 2015-2016. Available at: <u>www.cpacanada.ca/en/the-cpa-profession/about-cpa-canada/annual-reports/2015-2016</u>.
- Croson, R., and U. Gneezy. 2009. Gender differences in preferences. *Journal of Economic Literature* 47: 448-74.

- Dalton, D. R., J. W. Hill, and R. J. Ramsay. 1997. Women as managers and partners: Context specific predictors of turnover in international public accounting firms. *Auditing: A Journal of Practice & Theory* 16: 29-50.
- Dalton, D. W., J. R. Cohen, N. L. Harp, and J. J. Mcmillan. 2014. Antecedents and consequences of perceived gender discrimination in the audit profession. *Auditing: A Journal of Practice* & *Theory* 33: 1-32.
- Dambrin, C., and C. Lambert. 2008. Mothering or auditing? The case of two Big Four in France. *Accounting, Auditing and Accountability Journal* 21: 474-506.
- de Accountant. 2010. *Cultuur, cultuur en nog eens cultuur*. Available at: <u>www.accountant.nl/globalassets/accountant.nl/blad/2010-nr.-</u> <u>3/acc_2010_3_cultuur_cultuur_en_nog_eens_cultuur.pdf</u>
- Dodgson, M. K., C. P. Agoglia, G. B. Bennett, and J. R. Cohen. 2020. Managing the auditorclient relationship through partner rotations: The experiences of audit firm partners. *The Accounting Review* 95(2): 89-111.
- Fernandez-Mateo, I., and R. M. Fernandez. 2016. Bending the pipeline? Executive search and gender inequality in hiring for top management jobs. *Management Science* 62: 3636-55.
- FRC (Financial Reporting Council). 2017. *Key facts and trends in the accountancy profession*. Available at: <u>www.frc.org.uk/getattachment/77fc8390-d0d1-4bfe-9938-8965ff72b1b2/Key-</u> Facts-and-Trends-2017.pdf.
- Gallhofer, S., C. Paisey, C. Roberts, and H. Tarbert. 2011. Preferences, constraints and worklifestyle choices: The case of female Scottish chartered accountants. *Accounting, Auditing and Accountability Journal* 24: 440-70.
- Gayle, G. L., L. Golan, and R. A. Miller. 2012. Gender differences in executive compensation and job mobility. *Journal of Labor Economics* 30: 829-72.
- Goldin, C., and L. F. Katz. 2008. Transitions: Career and family life cycles of the educational elite. *American Economic Review* 98: 363-69.
- Gorman, E. H. 2005. Gender stereotypes, same-gender preferences, and organizational variation in the hiring of women: Evidence from law firms. *American Sociological Review* 70: 702-28.
- Hardies, K., D. Breesch, and J. Branson. 2015. The female audit fee premium. *Auditing: A Journal of Practice & Theory* 34: 171-95.

- Hardies, K., D. Breesch, and J. Branson. 2016. Do (fe)male auditors impair audit quality? Evidence from going-concern opinions. *European Accounting Review* 25: 7-34.
- Hardies, K., M.-L. Vandenhaute, and B. Breesch. 2018. An analysis of auditors' going-concern reporting accuracy in private firms. *Accounting Horizons* 32: 117-32.
- Hay, D. C., W. R. Knechel, and N. Wong. 2006. Audit fees: A meta-analysis of the effect of supply and demand attributes. *Contemporary Accounting Research* 23: 141-91.
- Hultin, M., and R. Szulkin. 1999. Wages and unequal access to organizational power: An empirical test of gender discrimination. *Administrative Science Quarterly* 44: 453-72.
- Institute for The Equality of Women and Men. 2013. *The gender pay gap in Belgium: Report* 2013. <u>http://igvm-iefh.belgium.be/sites/default/files/downloads/63</u> Gender Pay Gap Report 2013.pdf. Retrieved September 12, 2017.
- Ittonen, K., K. Johnstone, and E.-R. Myllymäki. 2015. Audit Partner Public-Client Specialisation and Client Abnormal Accruals. *European Accounting Review* 24: 607-33.
- Ittonen, K., E. Vähämaa, and S. Vähämaa. 2013. Female auditors and accruals quality. *Accounting Horizons* 27: 205-28.
- Ittonen, K., and E. Peni. 2012. Auditor's gender and audit fees. *International Journal of Auditing* 16: 1-18.
- Jonnergård, K., A. Stafsudd, and U. Elg. 2010. Performance evaluations as gender barriers in professional organizations: A study of auditing firms. *Gender, Work and Organization* 17: 721-47.
- *Kassman v. KPMG LLP.* 2011. United States District Court Southern District of New York. Complaint filed on June 2.
- Knechel, W. R., L. Niemi, and M. Zerni. 2013. Empirical evidence on the implicit determinants of compensation in Big 4 audit partnerships. *Journal of Accounting Research* 51: 349-87.
- Komori, N. 2008. Toward the feminization of accounting practice. *Accounting, Auditing and Accountability Journal* 21: 507–38.
- Kornberger, M., C. Carter, and A. Ross-Smith. 2010. Changing gender domination in a Big Four accounting firm: Flexibility, performance and client service in practice. Accounting, Organizations and Society 35: 775-91.
- Kothari, S. P., A. J. Leone, and C. E. Wasley. 2005. Performance matched discretionary accrual measures. *Journal of Accounting and Economics* 39(1): 163-97.

- Kumra, S., and S. Vinnicombe. 2008. A study of the promotion to partner process in a professional services firm: How women are disadvantaged. *British Journal of Management* 19: S65-S74.
- Kurtulus, F. A., and D. Tomaskovic-Devey. 2012. Do female top managers help women to advance? A panel study using EEO-1 records. *Annals of the American Academy of Political and Social Science* 639: 173-97
- Lazear, E. P., and S. Rosen. 1990. Male-female wage differentials in job ladders. *Journal of Labor Economics* 8: S106-S123.
- Lee, H. S., A. Nagy, and A. Zimmerman. 2019. Audit partner assignments and audit quality in the United States. *The Accounting Review* 92: 297-323.
- Madsen, P. 2013. The integration of women and minorities into the auditing profession since the civil rights period. *The Accounting Review* 88: 2145-77.
- Matsa, D. A., and A. R. Miller. 2011. Chipping away at the glass ceiling: Gender spillovers in corporate leadership. *American Economic Review* 101: 635-39.
- Mueller, G., and E. J. S. Plug. 2006. Earnings effect of personality. ILR Review 60: 3-22.
- Nekby, L., P. S. Thoursie, and L. Vahtrik. 2008. Gender and self-selection into a competitive environment: Are women more overconfident than men? *Economics Letters* 100: 405-407.
- Niederle, M., and L. Vesterlund. 2007. Do women shy away from competition? Do men compete too much? *Quarterly Journal of Economics* 122: 1067-1101.
- Oaxaca, R. L., and M. R. Ransom. 1994. On discrimination and the decomposition of wage differentials. *Journal of Econometrics* 61: 5-21.
- Ooghe, H., and C. Spaenjers. 2005. De FiTO-meter: Een nieuwe, eenvoudige en geïntegreerde maatstaf voor de financiële toestand van een onderneming. *Accountancy & Bedrijfskunde* 25: 5-14.
- *Page v. PricewaterhouseCoopers LLP.* 2004. In the United States District Court for the Northern District of Illinois Eastern Division. Complaint filed on June 21.
- Phelps, E. S. 1972. The statistical theory of racism and sexism. *American Economic Review* 62: 659–661.
- *Price Waterhouse v. Hopkins.* 1989. In the Supreme Court of the United States (490 U.S. 228). Case decided on May 1.

- Reuben, E., P. Rey-Biel, P. Sapienza, and L. Zingales. 2012. The emergence of male leadership in competitive environments. *Journal of Economic Behavior & Organization* 83: 111-17.
- Saccardo, S., A. Pietrasz, and U. Gneezy. 2018. On the size of the gender difference in competitiveness. *Management Science* 64: 1541-54.
- Schaefer, J., and M. Zimmer. 1995. Gender and earnings of certain accountants and auditors: A comparative study of industries and regions. *Journal of Accounting & Public Policy* 14: 265-91.
- Schiffel, L., D. L. Schroeder, and K. A. Smith. 2013. IMA 2012 salary survey: Leaning into the wind. *Strategic Finance* 94: 27-45.
- Shipman, J., Q. Swanquist, and R. Whited. 2017. Propensity score matching in accounting research. *The Accounting Review* 92: 213-44.
- Stillabower, L. 1985. An evaluation of career issues for women CPAs. *The Ohio CPA Journal* Winter: 21-24.
- Sutter, M., and D. Glätzle-Rützler. 2015. Gender differences in the willingness to compete emerge early in life and persist. *Management Science* 61: 2339-54.
- Trapp, M. W., R. H. Hermanson, and D. H. Turner. 1989. Current perceptions of issues related to women employed in public accounting. *Accounting Horizons* 3: 71-85.
- Vandenhaute, M.-L., K. Hardies, and D. Breesch. 2020. Professional and commercial incentives in audit firms: Evidence on partner compensation. *European Accounting Review* 29: 521-54.
- Vanstraelen, A., and M. Willekens. 2008. Audit regulation in Belgium: Overregulation in a limited capital market oriented country? In *Auditing, Trust and Governance: Developing Regulation in Europe*, edited by R. Quick, S. Turley, and M. Willekens, 19-41. London, U.K.: Routledge.
- Whiting, R. H., E. Gammie, and K. Herbohn. 2015. Women and the prospects for partnership in professional accountancy firms. *Accounting and Finance* 55: 575–605.

APPENDIX Variable definitio	ns
FEMALE	= 1 if the audit engagement partner is female; = 0 if male.
OMALE%	= Percentage of male audit partners within the audit office.
LAF	= Natural logarithm of audit fees.
PRESTIGE	= 1 if the company is publicly traded, or the subsidiary of a publicly traded company, or in the top 10% of the office's largest clients (measured by total assets); = 0 otherwise.
LTA	= Natural logarithm of total assets.
LNSALES	= Natural logarithm of sales.
LIST	= 1 if the company is publicly traded; = 0 otherwise.
LEV	= Total liabilities divided by total assets.
ROA	= Net income divided by total assets.
CATA	= Current assets divided by total assets.
QUICK	= Current assets (less inventories) divided by current liabilities.
QUICK_RANK	= the rank-transformed value of <i>QUICK</i> .
IRISK	= Inventory and receivables divided by total assets.
LOSS	= 1 if the company reports a loss; = 0 otherwise.
PROBF	= Bankruptcy score, where lower values indicate that the company is at greater risk of bankruptcy (Ooghe and Spaenjers 2005).
LNAS	= Natural logarithm of non-audit fees.
BUSY	= 1 if the company has a December year-end; = 0 otherwise.
OPINION	= 1 if the company receives a modified audit opinion; = 0 otherwise.
BIG4	= 1 if the company is audited by a Big 4 firm; = 0 otherwise.
SPECF	= 1 if the company's audit firm is an industry specialist; = 0 otherwise. Audit firms are classified as specialists if they have the largest or second-largest market share, by total assets, in the industry (NACE two-digit codes).
SPECAP	= 1 if the audit engagement partner is an industry specialist; = 0 otherwise. Audit partners are classified as specialists if they have the largest or second-largest market share, by total assets, in the industry (NACE two-digit codes).
PUBSPEC	= 1 if the audit engagement partner is a public company specialist; = 0 otherwise. Audit partners are classified as public company specialists if they audit three or more publicly traded clients during the year (Ittonen et al. 2015).
LNCAREER	= Natural logarithm of the number of years since the auditor's certification date.
LNCLIENTS	= Natural logarithm of the number of clients of an engagement partner.
LANG	= 1 if the audit engagement partner has a French-speaking affiliation; = 0 otherwise.
ABFEE	= Aggregated abnormal audit fees at the partner-year level. Calculated as the residuals from an engagement-level audit fee regression (Col. 2 of Table 5 but without the

	<i>FEMALE</i> variable), where the residuals are then aggregated for each partner-year observation.
N_SPECF	= Total number of clients in audit partner i 's portfolio for which the audit firm was classified as an industry specialist in year t .
N_SPECAP	= Total number of clients in audit partner i 's portfolio for which the audit partner was classified as an industry specialist in year t .
OFFICE_SIZE	= Size of the local audit office as measured by the number of clients.
MEAN_LTA	= The mean of the natural log of total assets for partner <i>i</i> 's clients in year t .
MEAN_LEV	= The mean ratio of liabilities to assets for partner i 's clients in year t .
MEAN_ROA	= The mean return on assets for partner i 's clients in year t .
FEES_PARTNER	= Sum of audit fees of all partner i 's clients in year t .
Ln(COMP)	= Natural logarithm of audit partner compensation.

Female ability distribution and a higher promotion threshold for female partners



The female auditors who clear the higher promotion threshold have high ability.

Male ability distribution and a lower promotion threshold for male partners



Figure 1. The effect of gender discrimination on the observed abilities of female and male partners.

TABLE I		
Summary of gender discrimination hy	potheses and alternative theories	
	Effect of partner gender on fees and	How do the effects differ in offices that have more
Theory	client assignments (H1a and H2a)	male partners? (H1b and H2b)
1. Employer discrimination	H1a: Female partners are associated with fee premiums.	H1b: Female fee premiums are larger in male- dominated offices.
	H2a: Prestigious clients are assigned to male partners.	H2b: The association between prestigious clients and gender is larger in male-dominated offices.
2. Client preferences Clients, especially prestigious clients,	Male partners are associated with fee premiums (opposite to H1a).	Male fee premiums are larger in male-dominated offices (opposite to H1b).
prefer male partners.	Prestigious clients are assigned to male partners (consistent with H2a).	The association between prestigious clients and gender is larger in male-dominated offices (consistent with H2b).
3. Males are more productive	Male partners are associated with fee premiums (opposite to H1a). Prestigious clients are assigned to male partners (consistent with H2a).	No effect is expected (opposite to H1b and H2b).
4. Females are more productive	Female partners are associated with fee premiums (consistent with H1a). Prestigious clients are assigned to female partners (opposite to H2a).	No effect is expected (opposite to H1b and H2b).
5. Labor supply Offices have more males when there	No effect is expected (opposite to H1a and H2a).	Male fee premiums are larger in male-dominated offices (opposite to H1b).
is a larger supply of talented males.		The association between prestigious clients and gender is larger in male-dominated offices (consistent with H2b).

TADLE 1

TABLE 1 (con	it.)
--------------	------

Summary of gender discrimination hypotheses and alternative theories

6. Self-selection Male-dominated offices attract women with similar characteristics to men.	No effect is expected (opposite to H1a and H2a).	The association between fees and gender is smaller in male-dominated offices (opposite to H1b). The association between prestigious clients and gender is smaller in male-dominated offices (opposite to H2b).
7. Affirmative action Offices have more female partners when they discriminate in favor of women.	Male partners are associated with fee premiums (opposite to H1a). Prestigious clients are assigned to male partners (consistent with H2a).	Fee differentials between males and females are larger in gender-balanced offices (opposite to H1b). The association between prestigious clients and gender is larger in male-dominated offices (consistent with H2b).

TABLE 2Derivation of Sample

Description

Firm-Year Observations 2008–2014	127,076
Less Observations with More Than One Auditor (Joint-Audits)	(4,641)
Less Observations from Sole Practitioners	(11,288)
Less Observations with Missing Data for Audit Fees ^a	(5,953)
Less Financial Institutions and Public Administrative Institutions ^b	(10,128)
Less Observations with Missing Data for Control Variables	(184)
Firm-Year Observations for Final Sample	94,882

^a These data are missing despite the fact that companies are required by law to disclose audit fees in the notes to their financial statements.

^b Financial institutions are excluded because of their specific accounting requirements, which differ substantially from those of industrial and commercial firms. Public administrative institutions are excluded because of their specific nature.

TABLE 3

Descriptive statistics

Panel A: Descriptive statistics for the clients of female and male partners (N = 94,882 audit engagements).

		Tota	al Sample		Femal	e Partners		Male	e Partners	t-stat.
	(N =	94,882 enga	gements)		()	n = 9,907)		(<i>n</i> :	= 84,975)	(1) - (2)
Variable	Mean	Std. Dev.	Median	Mean	Std. Dev.	Median	Mean	Std. Dev.	Median	
LAF	8.867	1.032	8.780	8.930	1.011	8.868	8.860	1.034	8.774	6.39***
PRESTIGE	0.130	0.337	0.000	0.106	0.308	0.000	0.133	0.340	0.000	-7.46***
LTA	15.715	2.040	15.752	15.641	1.927	15.725	15.724	2.052	15.755	-3.84***
LNSALES	13.921	5.298	15.653	14.021	5.107	15.675	13.910	5.320	15.652	1.98^{**}
LEV	0.717	0.958	0.628	0.755	1.043	0.634	0.712	0.947	0.627	4.24^{***}
ROA	0.043	0.227	0.046	0.042	0.236	0.045	0.044	0.225	0.046	-0.61
CATA	0.675	0.328	0.798	0.699	0.323	0.834	0.673	0.329	0.793	7.26^{***}
QUICK	23.412	146.963	1.208	15.975	112.723	1.190	24.279	150.425	1.210	-5.32***
QUICK_RANK	47,294	27,389	47,295	47,162	26,937	46,464	47,310	27,441	47,393	-0.51
IRISK	0.508	0.325	0.538	0.532	0.328	0.571	0.505	0.324	0.533	7.74^{***}
LOSS	0.296	0.456	0.000	0.307	0.461	0.000	0.295	0.456	0.000	2.49^{***}
PROBF	0.553	0.071	0.564	0.552	0.073	0.563	0.553	0.071	0.564	-2.02^{**}
LIST	0.006	0.078	0.000	0.005	0.073	0.000	0.006	0.079	0.000	-1.12
LNAS	2.383	3.860	0.000	2.221	3.750	0.000	2.402	3.872	0.000	-4.43***
BUSY	0.849	0.358	1.000	0.809	0.393	1.000	0.854	0.353	1.000	-11.95***
OPINION	0.251	0.434	0.000	0.262	0.440	0.000	0.250	0.433	0.000	-2.75^{***}
BIG4	0.519	0.500	1.000	0.549	0.498	1.000	0.515	0.500	1.000	6.40^{***}
SPECF	0.283	0.450	0.000	0.282	0.450	0.000	0.283	0.450	0.000	-0.01
SPECAP	0.044	0.205	0.000	0.022	0.146	0.000	0.046	0.210	0.000	-11.29***
PUBSPEC	0.043	0.202	0.000	0.027	0.162	0.000	0.044	0.206	0.000	-8.18***
CAREER	17.073	7.085	17.000	14.986	5.777	15.000	17.316	7.183	17.000	-31.14***
LNCAREER	2.801	0.469	2.890	2.691	0.435	2.773	2.814	0.471	2.890	-24.79***
CLIENTS	76.737	55.285	64.000	63.717	40.973	55.000	78.255	56.524	65.000	-24.85***
LNCLIENTS	4.047	0.849	4.159	3.867	0.870	4.007	4.068	0.844	4.174	-22.41***
LANG	0.264	0.441	0.000	0.308	0.462	0.000	0.258	0.438	0.000	10.62***

ranel D: Descriptive s	statistics solied by (JMALE % (IN = 92)	+,002 audit eligage	ements).			
					(2)		
	Offices with a	i more diverse gei	nder balance		Male-domir	nated offices	t-stat.
		$0.000 \le OMAL$	<i>E</i> % < 0.750		$0.750 \le OMAL$	$E\% \le 1.000$	(1) - (2)
			(n = 14,908)			(n = 79,974)	
Variable	Mean	Std. Dev.	Median	Mean	Std. Dev.	Median	
LAF	8.832	0.949	8.743	8.874	1.046	8.780	-4.51***
PRESTIGE	0.122	0.327	0.000	0.132	0.338	0.000	-3.22***
LTA	15.750	1.931	15.808	15.709	2.059	15.741	2.25^{**}
LNSALES	14.250	4.963	15.799	13.860	5.356	15.627	8.25^{***}
LEV	0.704	0.908	0.625	0.719	0.967	0.628	-1.85*
ROA	0.043	0.219	0.044	0.043	0.228	0.046	0.06
CATA	0.678	0.321	0.796	0.675	0.329	0.799	1.05
QUICK	21.224	140.544	1.185	23.820	148.126	1.213	-1.98**
QUICK_RANK	46,933	26,734	46,365	47,536	27,510	47,659	-2.47***
IRISK	0.515	0.320	0.546	0.507	0.325	0.536	2.80^{***}
LOSS	0.290	0.454	0.000	0.297	0.457	0.000	-1.59
PROBF	0.554	0.070	0.564	0.553	0.071	0.564	2.05^{**}
LIST	0.005	0.072	0.000	0.006	0.080	0.000	-1.62
LNAS	2.108	3.674	0.000	2.434	3.892	0.000	-9.47***
BUSY	0.839	0.368	1.000	0.851	0.356	1.000	-4.04***
OPINION	0.240	0.427	0.000	0.253	0.435	0.000	-3.21***
BIG4	0.382	0.486	0.000	0.544	0.498	1.000	-36.73***
SPECF	0.188	0.391	0.000	0.300	0.458	0.000	-28.08^{***}
SPECAP	0.030	0.169	0.000	0.047	0.211	0.000	-9.37***
PUBSPEC	0.041	0.198	0.000	0.043	0.203	0.000	-1.11
LNCAREER	2.856	0.464	2.944	2.790	0.469	2.890	-15.66***
LNCLIENTS	3.847	0.822	3.951	4.085	0.849	4.190	-31.62***
LANG	0.318	0.466	0.000	0.253	0.435	0.000	-16.56***
$^{*, **, ***} n \leq .10, .05, .01$	l. respectively.						

TABLE 3 (cont.) **Panel B:** Descriptive statistics sorted by OMALF% (N = 94 882 audit engagements)

TABLE 4												
Correlation Matrix	(N = 94,88)	2).										
Correlations that ar	e statistical	ly signific	ant at the 9	95% level	(two-tailed	l) are high	lighted in l	old. All v	ariables ar	e defined	in the Apr	pendix.
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
1. FEMALE	1											
2. OMALE%	-0.465	1										
3. LAF	0.021	-0.049	1									
4. PRESTIGE	-0.024	-0.005	0.271	1								
5. LTA	-0.013	-0.026	0.581	0.491	1							
6. LNSALES	0.006	-0.011	0.478	0.073	0.399	1						
7. LEV	0.014	-0.000	-0.065	-0.056	-0.336	-0.138	1					
8. ROA	-0.002	0.005	0.059	0.010	0.197	0.174	-0.386	1				
9. CATA	0.024	-0.013	0.034	-0.168	-0.244	-0.099	0.056	0.057	1			
10. QUICK	-0.017	-0.002	-0.134	0.021	-0.030	-0.284	-0.094	-0.014	0.045	1		
11. IRISK	0.025	-0.004	0.076	-0.118	-0.092	0.185	0.002	0.070	0.759	-0.007	1	
12. LOSS	0.008	-0.003	-0.071	-0.044	-0.180	-0.230	0.233	-0.513	-0.121	-0.008	-0.108	1
13. PROBF	-0.007	0.087	0.087	0.016	0.249	0.348	-0.510	0.659	0.016	-0.016	-0.021	-0.609
14. LIST	-0.004	0.003	0.138	0.204	0.100	0.013	-0.016	-0.009	-0.070	-0.011	-0.057	0.007
15. LNAS	-0.014	0.023	0.365	0.149	0.250	0.153	-0.027	0.008	-0.043	-0.044	-0.020	-0.010
16. BUSY	-0.039	0.034	-0.072	0.024	-0.024	-0.034	-0.007	-0.013	-0.040	0.001	-0.023	0.028
17. OPINION	0.009	-0.004	-0.051	-0.049	-0.143	-0.090	0.249	-0.262	-0.023	-0.024	-0.006	0.343
18. BIG4	0.021	-0.010	0.273	0.010	0.072	-0.091	0.029	-0.025	0.016	0.052	-0.010	0.056
19. SPECF	-0.000	0.018	0.180	0.056	0.088	-0.051	0.008	-0.011	-0.006	0.038	-0.026	0.034
20. SPECAP	-0.037	0.028	0.092	0.131	0.131	-0.015	0.006	-0.017	-0.082	0.027	-0.066	0.029
21. PUBSPEC	-0.027	-0.007	0.008	0.092	0.032	-0.044	0.012	-0.013	-0.050	0.017	-0.042	0.029
22. LNCAREER	-0.080	-0.014	-0.078	0.010	0.009	0.017	-0.014	0.011	-0.023	-0.017	0.025	0.008
23. LNCLIENTS	-0.073	0.091	-0.002	-0.018	-0.019	-0.066	0.027	-0.015	-0.001	0.010	0.017	0.038
24. LANG	0.034	-0.051	0.013	0.024	0.008	-0.038	0.028	-0.015	-0.040	0.003	-0.039	0.028

TABLE 4 (cont.)												
Correlation Matrix ($N = 94,882$).												
	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
13. PROBF	1										(-)	
14. LIST	-0.015	1										
15. LNAS	0.016	0.111	1									
16. BUSY	-0.025	0.013	-0.023	1								
17. OPINION	-0.375	-0.005	-0.017	-0.009	1							
18. BIG4	-0.053	0.006	0.077	-0.007	-0.013	1						
19. SPECF	-0.033	0.025	0.072	-0.001	-0.005	0.554	1					
20. SPECAP	-0.025	0.044	0.051	0.026	-0.001	0.126	0.254	1				
21. PUBSPEC	-0.034	0.111	-0.000	0.030	-0.005	0.008	0.015	0.032	1			
22. LNCAREER	-0.007	-0.004	-0.014	-0.039	-0.009	0.021	-0.137	-0.012	-0.027	1		
23. LNCLIENTS	-0.035	-0.011	-0.018	0.025	0.045	0.394	0.221	0.089	0.060	-0.073	1	
24 LANG	-0.041	0.018	-0.053	0.030	-0.020	-0.053	-0.039	0.039	0.075	-0.035	-0.023	1

TABLE 5

Audit fees, partner gender, and the gender composition of the partner's office (H1a and H1b). Dependent Variable = LAF (log of audit fees).

I	Mo	<u>del 1</u>	Mo	del 2	Mo	del <u>3</u>	Mo	del 4
	Coefft.	t-stat.	Coefft.	t-stat.	Coefft.	t-stat.	Coefft.	t-stat.
FEMALE	0.039	2.96^{***}	0.032	2.41^{**}	-0.352	-8.32^{***}	-0.260	-5.61***
FEMALE × OMALE%					0.491	8.71^{***}	0.371	5.95***
OMALE%					-0.295	-9.58^{***}	-0.232	-7.03^{***}
LTA	0.238	64.26***	0.237	64.20^{***}	0.237	64.11***	0.236	64.13***
LNSALES	0.053	43.62***	0.054	44.14^{***}	0.053	43.73***	0.054	44.17^{***}
LEV	0.087	12.94***	0.087	12.99***	0.086	12.88^{***}	0.086	12.93***
ROA	0.041	1.74^*	0.033	1.39	0.041	1.73^{*}	0.032	1.37
CATA	0.418	16.37***	0.394	15.52^{***}	0.414	16.29***	0.394	15.55^{***}
QUICK_RANK	0.000	0.04	-0.000	-0.10	-0.000	-0.01	-0.000	-0.16
IRISK	-0.194	-9.00^{***}	-0.185	-8.65^{***}	-0.192	-8.94^{***}	-0.185	-8.65^{***}
LOSS	0.058	6.65^{***}	0.055	6.29^{***}	0.058	6.67^{***}	0.055	6.31***
PROBF	-1.128	-10.56***	-1.087	-10.24^{***}	-1.122	-10.54^{***}	-1.087	-10.26***
LIST	0.892	13.47***	0.890	13.31***	0.893	13.36***	0.891	13.25***
LNAS	0.046	42.62***	0.044	41.23***	0.046	42.86***	0.044	41.39***
BUSY	-0.095	-7.64^{***}	-0.094	-7.64^{***}	-0.093	-7.47^{***}	-0.093	-7.56^{***}
OPINION	-0.001	-0.14	-0.013	-1.47	-0.004	-0.42	-0.014	-1.60
BIG4	0.547	49.03***	0.476	40.50^{***}	0.533	47.72^{***}	0.471	39.92***
SPECF	0.004	0.35	-0.001	-0.07	0.004	0.38	-0.001	-0.01
SPECAP	0.004	0.17	0.002	0.07	0.006	0.26	0.003	0.11
PUBSPEC	0.050	2.65^{***}	0.035	1.87^*	0.050	2.64^{***}	0.041	2.18^{**}
LNCAREER	-0.024	2.92^{***}	-0.037	-4.41***	-0.028	-3.36***	-0.039	-4.61***
LNCLIENTS	-0.082	-15.60^{***}	-0.073	-13.29***	-0.078	-14.76^{***}	-0.071	-12.97^{***}
LANG	0.077	7.61***	0.024	1.59	0.071	7.04***	0.019	1.28
Observations	94	,882	94	,882	94	,882	94,	882
Year fixed effects	Y	Zes	Y	les	Y	es	Y	es
Industry fixed effects	Y	es	Y	es	Y	es	Y	es
City fixed effects	l	No	Y	es	1	No	Y	es
Adjusted R ²	0	.60	0	.60	0	.60	0.	60

*, **, *** represent significance levels of 0.10, 0.05, and 0.01, respectively. All tests are two-tailed. Standard errors are adjusted for heteroscedasticity and clustered on each company.

TABLE 6

The aggregated abnormal audit fees of female and male partners (N = 3,468 partner-year observations). Aggregated abnormal audit fees are calculated as the residuals from the engagement-level audit fee regression and the residuals are then aggregated for each partner-year observation.

Panel A: Univariate results (the mean values represent aggregated abnormal audit fees).

Offices with a more d	iverse gender balance	Male-dominated offices				
$0.000 \le OMA$	LE% < 0.750	$0.750 \le OMALE\% \le 1.000$				
Male partner-years	Female partner-years	Male partner-years	Female partner-years			
(n = 382)	(n = 279)	(n = 2,626)	(n = 181)			
1.090	0.618	0.055	2.913			
t-stat. =	= -0.61	t-stat.	= 4.38			

Panel B: Regression results. Dependent Variable = *ABFEE*.

	Model 1 Model 2		del 2	Model 3		Model 4		
	Coefft.	t-stat.	Coefft.	t-stat.	Coefft.	t-stat.	Coefft.	t-stat.
FEMALE	1.622	3.17***	1.589	3.10**	-7.079	-3.95***	-5.524	-3.04^{***}
FEMALE × OMALE%					10.938	4.61***	9.074	3.75^{***}
OMALE%					-7.090	-5.12***	-5.928	-4.12^{***}
BIG4	0.829	1.13	1.321	1.79^{*}	0.585	0.80	1.142	1.55
N_SPECF	-0.169	-11.61**	*-0.168	-11.82^{**}	-0.161	-11.09^{***}	-0.162	-11.43^{***}
N_SPECAP	0.088	2.01^{*}	0.105	2.46^{**}	0.095	2.18^{**}	0.109	2.56^{**}
PUBSPEC	1.023	0.88	1.375	1.22	1.237	1.07	1.594	1.41
LNCAREER	-0.234	-0.77	-0.323	-1.07	-0.214	-0.71	-0.293	-0.97
LNCLIENTS	0.816	4.81***	1.045	6.13***	0.770	4.56^{***}	0.993	5.82^{***}
LANG	-0.220	-0.61	-2.292	-4.24***	-0.332	-0.92	-2.374	-4.40^{***}
OFFICE_SIZE	-0.000	-0.02	-0.005	-4.89***	-0.001	-0.66	-0.005	-5.20^{***}
MEAN_LTA	0.842	5.41***	0.743	4.85^{***}	0.814	5.25***	0.738	4.82^{***}
MEAN_LEV	0.000	3.27***	0.000	2.95^{***}	0.000	3.24***	0.000	2.94^{**}
MEAN_ROA	-0.001	-3.01***	-0.001	-3.00***	-0.001	-3.09***	-0.001	-3.06***
Observations	3,4	468	3,4	468	3,4	468	3,	468
Year fixed effects	Ŷ	es	Ŷ	es	Y	es	Y	Yes
City fixed effects	Ν	No	Y	es	Ν	lo	Y	Yes
Adjusted R^2	0.	.06	0.	.11	0.	06	0	.12

*, **, *** represent significance levels of 0.10, 0.05, and 0.01, respectively. All tests are two-tailed. Standard errors are adjusted for heteroscedasticity and clustered on each company.

TABLE 7

The annual compensation of male and female audit partners (1,343 partner-years).

Panel A: Univariate comparisons of raw and logged compensation for male and female partners Male partners Female partners

	Maie partiters	i cinale parti		
	(n = 1, 165)	(n = 178)		
COMP (Euros)	232,142	140,916		
Female vs Male	t-stat. =	-4.93		
Ln(COMP)	11.83	11.40		
Female vs Male	t-stat. =	= -4.89		

Panel B: Compensation regressions. Dependent Variable = *Ln*(*COMP*).

	Model 1		Model 2		Model 3		Model 4	
	Coefft.	t-stat.	Coefft.	t-stat.	Coefft.	t-stat.	Coefft.	t-stat.
FEMALE	-0.192	-2.81***	-0.252	-3.50^{***}	-0.617	-2.60^{***}	-1.151	-4.44***
$FEMALE \times$					0.535	1.72^{*}	1.227	3.59***
OMALE%								
OMALE%					-0.353	-1.84^{*}	-0.533	-2.45**
BIG4	0.350	3.62***	0.369	3.73***	0.338	3.49***	0.334	3.36***
N_SPECF	-0.002	-1.08	-0.000	-0.12	-0.001	-0.88	-0.000	-0.19
N_SPECAP	0.005	0.90	0.007	1.16	0.006	0.94	0.007	1.24
PUBSPEC	0.069	0.54	-0.011	-0.08	0.085	0.66	-0.014	-0.11
LNCAREER	0.197	3.97***	0.240	4.79^{***}	0.200	4.03***	0.255	5.09***
LNCLIENTS	-0.014	-0.29	-0.051	-0.99	-0.013	-0.26	-0.060	-1.18
LANG	-0.017	-0.34	-0.077	-1.12	-0.032	-0.64	-0.099	-1.43
FEES_PARTNER	0.259	5.95***	0.265	5.85***	0.255	5.84^{***}	0.266	5.89***
OFFICE_SIZE	0.001	5.37***	0.000	3.07***	0.000	5.24***	0.000	3.09**
MEAN_LTA	0.023	0.96	0.030	1.24	0.022	0.89	0.029	1.19
MEAN_LEV	0.000	1.13	0.000	0.97	0.000	1.12	0.000	0.88
MEAN_ROA	0.000	0.23	0.000	0.12	0.000	0.17	0.000	0.09
Observations	13	43	1 3	43	13	43	13	343
Year fixed effects	Ye	25	Y	es	Y	es	Y	es
City fixed effects	N	0	Ŷ	es	N	ίο Ο	Ŷ	es
Adjusted R^2	0.4	13	0.4	46	0.4	43	0.4	46

*, **, *** represent significance levels of 0.10, 0.05, and 0.01, respectively. All tests are two-tailed. Standard errors are adjusted for heteroscedasticity and clustered on each company.

TABLE 8

The gender of the audit engagement partner, client prestige, and the gender composition of the office (H2a and H2b). The models are estimated using logit. **Panel A:** Dependent Variable = FEMALE

Model 1		Model 2		
Coefft.	z-stat.	Coefft.	z-stat.	
-0.223	-3.14^{***}	0.597	2.68^{***}	
		-1.159	-2.44^{**}	
		-8.394	-48.86^{***}	
0.046	2.04^{**}	0.064	2.61**	
0.081	0.89	0.090	0.92	
0.004	0.04	-0.030	-0.24	
-0.000	-0.99	-0.000	-0.61	
0.283	2.93^{***}	0.351	3.23^{***}	
0.114	2.62^{***}	0.118	2.35^{**}	
0.484	1.22	0.382	0.88	
-0.013	-2.54^{**}	-0.001	-0.25	
-0.052	-1.08	0.610	9.61***	
94,	882	94,8	882	
Yes		Y	Yes	
Yes		Yes		
Yes Ye		es		
0.'	75	0.87		
0.13 0.		30		
	<u>Moc</u> Coefft. -0.223 0.046 0.081 0.004 -0.000 0.283 0.114 0.484 -0.013 -0.052 94,5 Y Y Y O.7 0.	$\begin{array}{c c} \underline{Model 1} \\ Coefft. & z-stat. \\ -0.223 & -3.14^{***} \\ \hline 0.046 & 2.04^{**} \\ 0.081 & 0.89 \\ 0.004 & 0.04 \\ -0.000 & -0.99 \\ 0.283 & 2.93^{***} \\ 0.114 & 2.62^{***} \\ 0.484 & 1.22 \\ -0.013 & -2.54^{**} \\ -0.052 & -1.08 \\ \hline 94,882 \\ Yes \\ O.75 \\ 0.13 \\ \end{array}$	$\begin{array}{c cccc} \underline{Model 1} & \underline{Model} \\ \hline Coefft. & z-stat. & Coefft. \\ -0.223 & -3.14^{***} & 0.597 \\ & & -1.159 \\ & & & -8.394 \\ \hline 0.046 & 2.04^{**} & 0.064 \\ \hline 0.081 & 0.89 & 0.090 \\ \hline 0.004 & 0.04 & -0.030 \\ -0.000 & -0.99 & -0.000 \\ \hline 0.283 & 2.93^{***} & 0.351 \\ \hline 0.114 & 2.62^{***} & 0.118 \\ \hline 0.484 & 1.22 & 0.382 \\ -0.013 & -2.54^{**} & -0.001 \\ -0.052 & -1.08 & 0.610 \\ \hline 94,882 & 94,8 \\ Yes & Yes & Yes \\ 0.75 & 0.8 \\ \hline 0.13 & 0.3 \\ \hline \end{array}$	

Panel B: Dependent Variable = PRESTIGE

	Model 1		Model 2	
	Coefft.	z-stat.	Coefft.	z-stat.
FEMALE	-0.264	-3.69^{***}	0.555	2.94^{***}
$FEMALE \times OMALE\%$			-0.360	-2.69^{***}
OMALE%			0.268	1.62
SPECAP	1.491	19.67***	1.492	19.69***
PUBSPEC	0.895	13.16***	0.902	13.25***
LNCAREER	0.059	1.46	0.057	1.42
LNCLIENTS	-0.107	-4.49^{***}	-0.107	-4.47^{***}
LANG	0.130	1.92^{*}	0.127	1.88^*
Observations	94,	882	94,	882
Year fixed effects	Yes		Yes	
Industry fixed effects	Yes		Yes	
City fixed effects	Yes		Yes	
Area under the ROC	0.69 0.69		69	
Pseudo R^2	0.	0.08 0.08		08

*, **, *** represent significance levels of 0.10, 0.05, and 0.01, respectively. All tests are twotailed. Standard errors are adjusted for heteroscedasticity and clustered on each company.